Author's response to reviews

Title: Potential for airborne transmission of infection in the waiting areas of healthcare premises: stochastic analysis using a Monte Carlo model

Authors:

   Clive B Beggs (c.b.beggs@bradford.ac.uk)
   Simon J Shepherd (s.j.shepherd@bradford.ac.uk)
   Kevin G Kerr (kevin.kerr@hdft.nhs.uk)

Version: 2 Date: 27 January 2010

Author's response to reviews:

Dear Dr Rajabi,

In response to your request, please find attached our revised manuscript, which now includes 'competing interests' and 'authors contribution' sections.

Regards,

Clive Beggs